Author's response to reviews

Title: A randomised control trial of the effectiveness of personalised letters sent subsequent to school dental inspections in increasing registration in unregistered children

Authors:

Chris J Cunningham (chris.cunningham@nhslothian.scot.nhs.uk)
Rob Elton (r.elton@ed.ac.uk)
Gail V A Topping (g.topping@chs.dundee.ac.uk)

Version: 4 Date: 6 February 2009

Author's response to reviews:

Dr Gail Topping
DHSRU
Kirsty Semple Way
Dundee
DD2 4BF

Tel: 07590010666

Dear Rikki

Many thanks for your email dated 6th February. I have uploaded the manuscript once again having checked the format and made some minor revisions to ensure it follows the format of the example that you sent (font and size of font on titles and format of references).

Could you please invoice DHSRU for the publication fee, this should be directed to Miss Hazel Braid as I understand that only Wellcome Trust publications are covered by the University of Dundee agreement with Biomed.

If there is anything else that you require of myself or my co-authors please do not hesitate to contact me.

Dr Gail Topping